Author's response to reviews

Title: Predictive value of coronary calcifications for future cardiac events in asymptomatic patients with diabetes mellitus: a prospective study in 716 patients over 8 years

Authors:

Alexander Becker (alexander.becker@med.uni-muenchen.de)
Alexander W Leber (alexander.leber@med.uni-muenchen.de)
Christoph Becker (christoph.becker@med.uni-muenchen.de)
Franz von Ziegler (franz.ziegler@med.uni-muenchen.de)
Janine Tittus (janine.tittus@med.uni-muenchen.de)
Ines Schroeder (ines.schroeder@med.uni-muenchen.de)
Gerhard Steinbeck (gerhard.steinbeck@med.uni-muenchen.de)
Andreas Knez (andreas.knez@med.uni-muenchen.de)

Version: 5 Date: 6 October 2008

Author's response to reviews: see over
Formatting changes for MS: 4749032322003579

Title: Predictive value of coronary calcifications for future cardiac events in asymptomatic patients with diabetes mellitus: a prospective study in 716 patients over 8 years


Dear Editor,

thank you for accepting the above-cited manuscript for publication. Below you will find the changes of the manuscript according to the formatting requirements:

1. The list of authors is now written exactly as they are in the submission system.
2. Alexander Becker is the corresponding author.
3. We have no acknowledgments to make.
4. Unnecessary capitals were removed from the title.
5. The authors’ qualifications were removed.
6. Germany is now included in the affiliation details.
7. Running title was removed.
8. Email addresses are now given on the title page.
9. Table legends and tables are now combined into a single section.

We hope these changes meet your requirements. Thank you for your help.

Yours sincerely,

A. Becker MD

Alexander Becker, MD
Department of Cardiology
Ludwig-Maximilians-University, Munich
D-81377 Munich
Germany
Tel: +49 89 7095-2229
Fax: +49 89 7095-2227
Email: alexander.becker@med.uni-muenchen.de